Author's response to reviews

Title: A new orthotic device to discourage recurrent dislocation of total hip replacement: Case report

Authors:

King Wong (kinghwong@hotmail.com)
Manoj Sivan (drmanojsivan@yahoo.com)
Gordon Matthews (mgm@doctors.org.uk)

Version: 2 Date: 5 October 2007

Author's response to reviews:

'case report' added to the title
section - abstract added to manuscript
'case report' changed to 'case presentation'
section - conclusion added
sections - consent, competing interests, authors contributions, acknowledgements added